Author’s response to reviews

Title: CYSTIC ANGIOMATOSIS: CASE REPORT OF PLEURAL EFFUSION AND MULTIPLE BONE LESIONS MIMICKING A METASTATIC MALIGNANT NEOPLASIA

Authors:

CAROLINE ANJOS (carolinesanjos@gmail.com)
RAFAELA BORGES (rafaela_meloborges@hotmail.com)
ANANDA CHAVES (chaves.ananda@gmail.com)
WILLIAM FUZITA (w_fuzita@yahoo.com.br)
CARLOS SILVA (augustomoreirasilva@yahoo.com.br)
UBIRATÃ OLIVEIRA (bira.glo@hotmail.com)

Version: 1 Date: 05 Feb 2019

Author’s response to reviews:

Revisor 1: The abstract was edited

Revisor 3: We agreed to delete Table 2 and important data was entered into the text.

Revisor 4: We revise vocabulary as directed.

Due to difficult access from the site of the bone lesion for biopsy and before the radiological signs, it was opted for not repeating the procedure. Because of the low specificity of pleural biopsy, the medical team opted for not carrying out the procedure.

The light criteria were introduced in Table 1

Additionally, alkaline phosphatase, PTH and D vitamin were normal